\documentclass[11pt]{article}

\include{lecture}

\begin{document}

\lecture{0}{9/6/2011}{Template}{Your name}

\draft % put this here only if you want to indicate this is a draft

\section{Environments}

\begin{theorem}
This is an example of the theorem environment.
Other theorem-like environments are: claim and exercise.
\end{theorem}
\begin{example}
This is an example of the example environment.
\end{example}

\section*{Acknowledgements}
This is an example of an acknowledgement section, which you can use at
the very end of the notes to thank the people who scribed the lectures
from previous years which you perused:

In writing the notes for this lecture, I perused the notes by VV for
lecture WW from the Spring 2007 offering of CS~810, and the notes by
XX for lecture YY from the Spring 2010 offering of CS~710.

Note that you will often need to refer to two lectures from the same
offering, as the partitioning may be different this year.

\end{document}
